Author's response to reviews

Title: A prognostic tool to identify adolescents at high risk of becoming daily smokers

Authors:

Igor Karp (igor.karp.1@umontreal.ca)
Gilles Paradis (gilles.paradis@mcgill.ca)
Marie Lambert (marie.lambert@umontreal.ca)
Erika Dugas (Erika.Dugas@crchum.qc.ca)
Jennifer O’Loughlin (jennifer.oloughlin@umontreal.ca)

Version: 5 Date: 2 August 2011

Author's response to reviews:

Please find attached the revised version of the manuscript, in which we have introduced all the requested modifications.